Author's response to reviews

Title: Ambient biomass smoke and cardio-respiratory hospital admissions in Darwin, Australia

Authors:

Fay H Johnston (fay.johnston@utas.edu.au)
Ross S Bailie (ross.bailie@menzies.edu.au)
Louis S Pilotto (l.pilotto@unsw.edu.au)
Ivan C Hanigan (ivan.hanigan@anu.edu.au)

Version: 3 Date: 23 August 2007

Author's response to reviews:

DESCRIPTION OF CHANGES

Minor content change in response to reviewer - In methods removal of the word 'passive' to describe the microVol aerosol sample - Done

Formatting changes
1. Title - Please remove the full stop from the end of the title. DONE
2. Highlighting/Tracking - Please remove any highlighting/tracking from your manuscript text. DONE
3. Introduction - Please may we suggest that the sub-heading 'Introduction' be removed - DONE
4. Section headings - Please use sentence case for all headings and sub-headings - DONE
5. Ethical approval A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate. - DONE
6. Manuscript sections - Please ensure that the remaining sections in the manuscript are in the correct order. - DONE
7. Funding - Please remove the sub-heading 'Funding'. Please include the statement of funding in the Acknowledgements section. - DONE
8. Competing interests - Please move this section to between the Conclusions and Authors' contributions. If there are none to declare, please write 'The authors declare that they have no competing interests.' - DONE
9. Authors' contributions - Please move this section to between the Competing interests and Acknowledgements - DONE
10. Acknowledgements - Please move this section to between the Authors’ contributions and References - DONE

11. References - References should be cited in the text in ascending numerical order, using numbers in square brackets, before the full stop or comma or semi-colon. - DONE

12. Reference 40 is entered incorrectly. Please provide the full citation details (i.e. volume number and page numbers). - DONE

13. Extracted references - Please ensure that the extracted references match the references in the manuscript.

14. Table and figure placement text - Please remove Table placement text from pages 7 and 8. Please remove figure placement text from pages 9 and 10. - DONE

15. Figure legends - Please move these to a position after the References section.

16. In the Figure legends section please type
'Figure 2' not 'Figures 2'
'Figure 3' not 'Figures 3'
DONE

17. Table title - Please remove the full stop from the end of the table title. - DONE